Author's response to reviews

Title: Horror in the echo lab: instructive cases to avoid further complications

Authors:

Albert Varga (vargaa@in2nd.szote.u-szeged.hu)
Giuliano Pingitore (king@ifc.cnr.it)
Ferenc Lakatos (vargaa@in2nd.szote.u-szeged.hu)
Riccardo Bigi (Riccardo.Bigi@unimi.it)
Rafael Paya (vargaa@in2nd.szote.u-szeged.hu)
Eugenio Picano (picano@ifc.cnr.it)

Version: 2  Date: 20 June 2005

Author's response to reviews:

To the Editor: please find enclosed the revised manuscript: Complications during pharmacological stress echocardiography: a video-case series by Varga A et al.
The paper was revised according to the recommendations of Reviewer 1 (in case 3 the type of the ECG abnormality was specified, the grammatical errors corrected, and the typos were corrected as the Reviewer 2 required. Changes, requested by the editorial office, have also added.